Author's response to reviews

Title: Cancer patterns among children of Turkish descent in Germany: a study at the German Childhood Cancer Registry

Authors:

Jacob Spallek (jacob.spallek@uni-bielefeld.de)
Claudia Spix (spix@imbei.uni-mainz.de)
Hajo Zeeb (zeeb@imbei.uni-mainz.de)
Peter Kaatsch (kaatsch@imbei.uni-mainz.de)
Oliver Razum (oliver.razum@uni-bielefeld.de)

Version: 4 Date: 2 May 2008

Author's response to reviews:

Dear Editor,

We made all the required formatting changes listed in core pre-accept email of Fri, 02 May.

Kind regards

Jacob Spallek